Author's response to reviews

**Title:** Cognitive intervention through a training program for picture-book reading in community-dwelling older adults: a randomized controlled trial

**Authors:**
Hiroyuki Suzuki (suzukihy@tmig.or.jp)
Masataka Kuraoka (mkuraoka@tmig.or.jp)
Masashi Yasunaga (yasunga9@tmig.or.jp)
Kumiko Nonaka (nonaka@tmig.or.jp)
Ryota Sakurai (r_sakurai@hotmail.co.jp)
Rumi Takeuchi (takeuchi@tmig.or.jp)
Yoh Murayama (yhoyho05@tmig.or.jp)
Hiromi Ohba (ohba500@tmig.or.jp)
Yoshinori Fujiwara (fujiwayo@tmig.or.jp)

**Version:** 2  
**Date:** 30 July 2014

**Author's response to reviews:** see over
July 30, 2014

Editor-in-Chief, *BMC Public Health*

MS: 1853855520137783

Research article

Cognitive intervention through a training program for picture-book reading in community-dwelling older adults: a randomized controlled trial

Dear Editor-in-Chief;

Thank you for your email of July 29, 2014, regarding our manuscript, “Cognitive intervention through a training program for picture-book reading in community-dwelling older adults: a randomized controlled trial”. I attach here our revised manuscript.

We have our revised manuscript according to your comments. Since our trial has followed the CONSORT 2010 explanation, we believe that our manuscript is ready for peer review.

Thank you in advance for your kind consideration of this paper.

We are looking forward to your reply.

Sincerely,
Hiroyuki Suzuki, PhD.

Research team for Social Participation and Community Health

Tokyo Metropolitan Institute of Gerontology

35-2 Sakae-cho, Itabashi-ku, Tokyo 173-0015, Japan

TEL: +81-3-3964-3241

suzukihy@tmig.or.jp